Author’s response to reviews

Title: Composition and Diversity of the Subgingival Microbiome and its Relationship with Age in Postmenopausal Women: an Epidemiologic Investigation

Authors:

Michael LaMonte (mlamonte@buffalo.edu)
Robert Genco (rjgenco@buffalo.edu)
Michael Buck (mjbuck@buffalo.edu)
Daniel McSkimming (dim@buffalo.edu)
Lu Li (lli59@buffalo.edu)
Kathleen Hovey (koreilly@buffalo.edu)
Chrostopher Andrews (candrews@buffalo.edu)
wei Zheng (zhengw@buffalo.edu)
Yijun Sun (yijunsun@buffalo.edu)
Amy Millen (aemillen@buffalo.edu)
Maria Tsompana (mtsompana@buffalo.edu)
Hailey Banack (hrbanack@buffalo.edu)
Jean Wactawski-Wende (jww@buffalo.edu)

Version: 3 Date: 03 Sep 2019

Author’s response to reviews:

3-September-2019

We thank the editor for providing continued opportunity to revise our manuscript for consideration of publication in BMC Oral Health. We have addressed each of the below requirements (highlighted in yellow in the revision).

1. Please confirm that the Ethics committee that approved the experimental protocols for the microbiome study, approved every aspect of your research detailed in this paper. Please include this information in your Ethics approval and consent to participate section. - DONE
2. Please include a statement in your Funding section describing the role of the funding body in the design of the study, the collection, analysis, and interpretation of data and in writing the manuscript. - DONE

3. Currently, the heading of Table 5 says 'Supplemental Table 3'. Please amend this. - DONE (thank you for catching this oversight)

4. Please include an Acknowledgements section in your Declarations where you acknowledge anyone who contributed towards the article who does not meet the criteria for authorship including anyone who provided professional writing services or materials. Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements section.

If there are no acknowledgements, please state “Not applicable”. - DONE

5. Please remove the response to reviewers document from the end of your manuscript file as this is no longer needed in the publication process. - DONE